Author’s response to reviews

Title: The long-term effects of adolescent pregnancies in a community in Northern Ghana on subsequent pregnancies and births of the young mothers

Authors:

Anne-Sophie Yussif (aanne@arcor.de)
Anyetei Lassey (anyetlass@gmail.com)
Gabriel Ganyaglo (gganyaglo@hotmail.com)
Eva Kantelhardt (eva.kantelhardt@uk-halle.de)
Heike Kielstein (heike.kielstein@medizin.uni-halle.de)

Version: 2 Date: 20 Oct 2017

Author’s response to reviews:

The authors make it very clear in the title and in the paper that the findings are based on a small, purposive sample and that while this is representative of the group surveyed, it cannot be extrapolated to the whole of Northern Ghana.

We added the limitation of the number of women enrolled in the study on page 13, last paragraph.

Furthermore, we now mention a generous travel grant by the FIDE AG e.V. (Women’s Health in International Development).